

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Angelucci, Manuela

Working Paper Love on the rocks: alcohol abuse and domestic violence in rural Mexico

IZA Discussion Papers, No. 2706

Provided in Cooperation with:

IZA – Institute of Labor Economics

Suggested Citation: Angelucci, Manuela (2007) : Love on the rocks: alcohol abuse and domestic violence in rural Mexico, IZA Discussion Papers, No. 2706, Institute for the Study of Labor (IZA), Bonn,

https://nbn-resolving.de/urn:nbn:de:101:1-20080410119

This Version is available at: https://hdl.handle.net/10419/34290

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

IZA DP No. 2706

Love on the Rocks: Alcohol Abuse and Domestic Violence in Rural Mexico

Manuela Angelucci

March 2007

Forschungsinstitut zur Zukunft der Arbeit Institute for the Study of Labor

Love on the Rocks: Alcohol Abuse and Domestic Violence in Rural Mexico

Manuela Angelucci

University of Arizona and IZA

Discussion Paper No. 2706 March 2007

IZA

P.O. Box 7240 53072 Bonn Germany

Phone: +49-228-3894-0 Fax: +49-228-3894-180 E-mail: iza@iza.org

Any opinions expressed here are those of the author(s) and not those of the institute. Research disseminated by IZA may include views on policy, but the institute itself takes no institutional policy positions.

The Institute for the Study of Labor (IZA) in Bonn is a local and virtual international research center and a place of communication between science, politics and business. IZA is an independent nonprofit company supported by Deutsche Post World Net. The center is associated with the University of Bonn and offers a stimulating research environment through its research networks, research support, and visitors and doctoral programs. IZA engages in (i) original and internationally competitive research in all fields of labor economics, (ii) development of policy concepts, and (iii) dissemination of research results and concepts to the interested public.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character. A revised version may be available directly from the author.

IZA Discussion Paper No. 2706 March 2007

ABSTRACT

Love on the Rocks: Alcohol Abuse and Domestic Violence in Rural Mexico

What causes alcohol abuse and domestic violence and how can we stop them? These behaviors have multiple determinants, making the effects of changes in wife's and husband's income ambiguous. This paper estimates the effects of exogenous changes in wife's and husband's income on husband alcohol abuse and alcohol-induced violence using new data from rural Mexico. A long-lasting 20 dollar monthly increase in wife income decreases husbands' alcohol abuse by 15% and aggressive behavior by 21%; the extra money increase the wife's freedom and security, is spent on individual and household goods, and it crowds out transfers from the husband only for 5% of the wives whose income increases. Alcohol abuse and violence are insensitive to short-term fluctuations in husband's income. These findings suggest that the wife uses her higher income to reduce the consumption of goods that lower her utility, that alcohol abuse responds more to changes in permanent than in temporary income, and that targeting women as recipients of micro-credit or of other welfare programs may have beneficial effects in reducing alcohol dependence and domestic violence.

JEL Classification: D13, I18, O12

Keywords: public health, household behavior, Mexico

Corresponding author:

Manuela Angelucci Department of Economics University of Arizona McClelland Hall 401 Tucson, AZ 85721-0108 USA E-mail: angelucm@eller.arizona.edu

1 Introduction

The World Health Organization (WHO) declared that interpersonal violence and harmful alcohol use are major challenges to global public health. These issues are especially severe in Latin American countries: in Mexico, 44% of women 15 and older and living with a partner were victim of physical, sexual, psychological, or economic violence in 2003; 70% of males aged 18 to 65 were habitual drinkers, and 40% of males in the same age range drank at least 1 to 3 times a month in 2002. There is also strong evidence of binge drinking: 34% of drinkers had between 5 and 24 drinks at once.¹

Domestic violence and alcohol abuse have strong detrimental effects on both the victims and the drinkers, as well as negative externalities for society as a whole. The victims of violence are affected psychologically, physically, and economically.² Children who witnessed violence become more likely to accept and perpetrate this behavior (Pollack, 2002). While there are no data on the cost of domestic violence in Mexico, estimates of victims' loss of productive capacity are at least 1.6% and 2% of GDP in Nicaragua and Chile (Morrison and Orlando, 1999).

Excessive drinking has been linked to a wide range of diseases, such as gastrointestinal complications, increased risks of some types of cancers, and high blood pressure. Alcohol intoxication also affects physical motor coordination; it is a major cause of automobile accidents, as well as having detrimental social and psychological consequences because of its addictive nature: the WHO estimates that alcohol is responsible for 4% of all years of health lost through disability or premature death worldwide (Krug et al., 2002); in Mexico, 24% of male drinkers aged 12 to 65 committed some alcohol-related crime in 2002.

Lastly, there is a strong link between alcohol abuse and aggressive behavior. In 2003, one in four cases of domestic violence in Mexico involved alcohol abuse (ENDIREH, 2004). The psychiatric literature, summarized in Leonard (2000, 2005), finds that alcohol causes violence. These effects are sizeable: in 1976 a 10% increase in beer tax reduced the likelihood of severe parental violence against children by 2.3 percentage points in the United States (Markowitz and Grossman, 1998); between 1989 and 1991, the elasticity of violence with respect to alcohol

¹The Mexican data on violence come from the Encuesta Nacional sobre la Dinamica de las Relaciones en los Hogares, or ENDIREH, a 2003 nationally representative survey on relationship dynamics within the household. The Mexican alcohol data are from the nationally representative 2002 Encuesta Nacional de Adicciones, or National Survey of Addictions, henceforth ENA2002.

²Most of the victims from the ENDIREH sample report some form of psychological damage; moreover, 8% of the victims had bruises, 1.2% cuts and burns, or lost teeth; 1.9% had a hemorrhage or bled; 1% had a fracture; 1% had to undergo surgery; 1.2% had a premature delivery or an abortion; 1.8% had other gynecological problems; 1.8% fainted. 6% had to stop working or studying, while 1.7% had financial losses.

consumption for U.S. college students varied between 0.3 and 0.6 for offenses such as property damage, trouble with authorities, being involved in a fight, and sexual abuse (Grossman and Markowitz, 2001); higher alcohol prices reduce violence against wives (Markovitz, 1999).^{3,4}

Understanding what causes alcohol abuse and aggressive behavior is both interesting in itself and essential to design effective policies. There are at least two competing explanations for domestic violence: 1) in the intrinsic value view, violence increases utility *per se*; 2) in the instrumental value view, violence is used instead as a tool to control the victim's resources. These two rationales may coexist. Tauchen, Witte, and Long (1991) model domestic violence as having both functions, and find an inverse relationship between the risk of violence and women's income. Bloch and Rao (2002), instead, consider violence as a mechanism to appropriate resources from the wife's family, and find that husbands from a sample of South Indian households from potter communities are more likely to beat wives who belong to rich families to extract more resources out of them. Alcohol abuse also may have an intrinsic or an instrumental value, increasing the consumer's utility either directly or through its un-inhibiting effects, or both.⁵

The effects of wife and husband's income on domestic violence and alcohol abuse differ according to which of the two motives dominate: the intrinsic value theory predicts that the consumption of these goods is a positive function of the husband's income (as he can "purchase" more of them) and a negative function of the wife's income. According to the instrumental value theory, instead, violence, and the alcohol intake needed to trigger it, depend positively on the wife's income and negatively on the husband's income.

Irrespective of the rationale for drinking and violence, higher wife income may favor the dissolution of abusive relationships, or reduce the level of violence and alcohol dependence in surviving relationships by making divorce a more credible threat: Farmer and Tiefenhalter (1996) show that battered wives may use shelter visits or police calls to signal their unwillingness to accept violence; using Canadian data, Bowlus and Seitz (2006) find that violent marriages end in divorce when the wife has the possibility to leave; Stevenson and Wolfers (2006) establish a negative relationship between the adoption of unilateral divorce law and domestic violence in the U.S.A..

How income affects alcohol abuse and violence determines what type of policies to imple-

 $^{^{3}}$ Markovitz estimates alcohol price elasticities of violence against wife smaller than -1, but she warns that her estimates may suffer from multicollinearity.

⁴Mehlum *et al.* (2006) suggest that there may be a link between the higher rye price and the decreased violence through lower consumption of beer in 19th century Bavaria.

⁵Obviously, there may be additional causes for alcohol abuse and domestic violence. For example, Pollak (2002) discusses the importance of the inter-generational transmission of violence.

ment. For example, targeting women as direct recipients of conditional cash transfers or of micro-credit programs, which is becoming the norm in developing countries, might either have positive spillover effects by discouraging husbands' abusive behavior, or perverse effects on women's wellbeing, if it makes them the victims of violence.

The goal of this paper is to understand the determinants of alcohol abuse and domestic violence by estimating the causal effect of wife's and husband's income on alcohol abuse and aggressive behavior for poor husbands in rural Mexico. I estimate these effects using data from a sample of indigent families in rural Mexico that was collected to evaluate the Mexican flagship welfare program Oportunidades, a conditional cash transfer program that transfers money to women provided they attend nutrition and health classes, their family has regular health checks, and their children attend school regularly.

Since the program was offered only to a random group of villages in my sample, I use program availability as instrument for wife's income I use temporary health shocks of household members, natural disasters, and village agricultural wage as instruments for husband's income. The previous empirical literature regressed violence on income without accounting for its likely endogeneity (e.g. Tauchen, Witte, and Long, 1991, and Bloch and Rao, 2002). On the other hand, Miguel (2005) shows how extreme rainfall - normally associated with sizeable income decreases - causes a surge in witch killings in Tanzania, but has no income data.

I find that a 20 dollar monthly transfer to the wife - a 35% increase in the income of both spouses - decreases alcohol abuse by 15%, and that this reduces aggressive behavior by 21%. This is consistent with the intrinsic view. The extra income is managed by the woman, increases her freedom and security, does not crowd out transfers from the husband, and it is spent on goods that increase both spouses' utility.

There are two main policy implications: first, increasing female income in poor rural Mexican households (through direct transfers, micro-credit programs, but also via better employment opportunities) is a useful tool to reduce alcohol dependence and eradicate domestic violence. Second, the success of a given policy crucially hinges on local cultural norms; a policy that might empower indigent Mexican women might harm South Indian wives.

The effect of husband's income is positive but statistically insignificant. Since the instruments for husband are temporary income shocks, this finding suggests that violence and alcohol abuse are insensitive to short-term income fluctuations, while they may depend on permanent income, which changes the relative bargaining power.

The rest of the paper has the following structure: Section 2 discusses the two contrasting theories of domestic violence and alcohol abuse, and their relationship with wife and husband's income. Section 3 describes the sample and compare its data on violence and alcohol abuse with nationally representative data. Section 4 and 5 shows the econometric model, the assumptions under which the parameters of interest are identified, and the estimation issues, while Section 5 presents the results, discussing their interpretation and providing robustness checks. Section 6 concludes.

2 Domestic violence and alcohol abuse: two alternative explanations

The incidence of domestic violence and alcohol abuse may vary depending on what causes the abuser to drink and behave aggressively. In this section I discuss two different theories for domestic violence and alcohol abuse; the theories have opposite predictions of the effects of wife's and husband's income on these behaviors.⁶

Consider violence and alcohol abuse as activities that provide intrinsic utility for the husband, while decreasing the wife's utility.⁷ Define these activities as v, and abstract from issues such as actual alcohol price, and endogenous marriage formation. Income is not pooled.⁸ Husband income is higher than wife income. The husband consumes v, paying for it with a monetary transfer to the wife, t.⁹ The transfer increases the wife's utility and decreases the husband's. In each period the spouses agree on a level of v and a transfer t such that both individuals are at least as well off in the marriage as outside of it. Once the relationship is severed, each spouse's utility depends only on the individual income. Assume that divorced spouses cannot get back together (or that the cost of forming a new relationship is high): in this case, a wife's endurance of husband's abuses is a function of both partners' current and permanent incomes. An increase in husband's income decreases his marginal utility of income, inducing him to trade off more money for violence and alcohol abuse. The wife accepts a higher level of these goods, as long as her marginal utility of income exceeds the marginal (dis)utility of violence and alcohol abuse. Conversely, an increase in wife's income reduces the level of v(and the transfer).

Changes in current and permanent income may have different effects on v and t. For

⁶From now on I will assume that the husband is the perpetrator and the wife the victim of violence. This is consistent with the existing empirical evidence.

⁷This model does not consider the costs of alcohol abuse for the husband (e.g. health, forgone earnings, etc.).

⁸Attanasio and Lechene (2002) reject the hypothesis that households pool income using these data. This is also a feature of household arrangements in other countries. For instance, Duflo and Udry (2004) provide evidence from Cote d'Ivoire that each spouse has direct control of parts of their earnings, and that different types of income are earmarked for different expenditures.

⁹In the case of alcohol, he is paying for her consent to drink.

example, suppose that the price of v (i.e. how much violence and alcohol abuse the husband can buy with a transfer of one peso) is a function of permanent income, while the quantity depends on current income. This is much like the Pareto weights in a households risk-sharing model depend on wealth, while the size of the intra-spousal transfers depends on the income realization. In this case, an increase in wife's permanent income reduces v in two ways: by increasing the price of v, and by reducing the marginal utility of the transfer. Instead, a temporary income increase has only the latter effect, but does not change the price of v.

Now, instead, consider v as having an instrumental value, i.e. violence is a means to control the wife's resources, and drunkenness facilitates aggressive behavior. While the threat of violence is sufficient with perfect information, if there is asymmetric information the husband uses violence to receive more money from the wife. Bloch and Rao (2002) use this idea to analyze the relationship between dowries and post-marriage monetary transfers from the wife to the husband family in India, where wife abuse is a common means to extract resources from the wife family.¹⁰ Their model predicts that an increase in husband's income reduces violence by decreasing his marginal utility from the wife's income. Conversely, an increase in wife's income causes higher violence by lowering her marginal utility of income and increasing the amount of money she is willing to transfer to her husband, hence making violence more attractive to him. A similar argument applies for alcohol abuse.¹¹

Increases in wife's income will also increase her outside opportunities, especially when the woman is financially dependent on her husband. A credible threat of divorce would lower the level and alcohol abuse and domestic violence irrespective of the determinants of the husband's behavior.

3 Sample characteristics and comparisons with nationally representative data

The data I use are a sample of poor households who live in 506 rural villages from the Mexican states of Guerrero, Hidalgo, Michoacan, Puebla, Queretaro, San Luis Potosi, and Veracruz. The primary purpose for the data collection was the evaluation of the conditional cash transfer program Oportunidades - the flagship welfare program of the Mexican government, started in 1998 and still ongoing, aimed at fostering the accumulation of human capital and at improving

¹⁰In their model, the money is actually obtained from the wife's family.

¹¹This perverse effect of higher wife's income on domestic violence was perceived as a potentially important issue by the Mexican policy makers when they where debating whether to have the wife or mother in the household as the direct recipient of the *Oportunidades* transfer.

nutrition and health. I have information on a variety of outcomes, including detailed income data and questions on alcohol abuse and related aggressive behavior.

From an initial sample of roughly 12,200 poor households in treatment and control villages, I excluded households where the wife is not the survey respondent, since the husband is less likely to disclose information on his violent behavior.¹² I further omitted single-headed households, households where multiple families lived in the same dwelling, and families with at least one eligible child who was not enrolled in school in the academic year before the program started (for reasons I will discuss later). The data I use in the empirical analysis are a sample of 10,257 households.

Qualitative village-level data collected in May 1999 reveal that alcohol abuse and domestic violence are perceived as important issues: 27% and 6% of respondents consider respectively alcoholism and domestic violence major problems in the sampled villages. The main dependent variables are the existence of habitual drinkers in the households, and the existence of drunken aggressive behavior. Household-level data on alcohol abuse and aggressive behavior are collected in November 1998, about 6 months after the beginning of the program. The questions asked are "Who is (are) the individual(s) who drinks the most in this household, irrespective of the frequency?" and "While drinking, does this person (referred to the heaviest drinker) have an aggressive behavior?"¹³ About 30% of the households have at least one drinker, which in 95% of the cases is the male household head. Very few households report having multiple drinkers. 70% of households report having no drinker. Since drinking is much more common in rural areas than what reported in my sample (about 61% of ENA2002 males aged 18 to 65 drink), the respondents most likely interpret this question as inquiring about alcohol abuses, or at least frequent consumption.¹⁴ This is consistent with the national evidence: in ENA2002, 36% of rural males aged 18 to 65 drink at least once a month¹⁵, while 26% are frequent drinkers (more than half of this group drinks once a week, and about 25% several times per week).

About 14% of drinkers in this sample behave aggressively after drinking. Thus, drunken violence occurs in 4.25% of the households. Despite keeping only households where the wife is the respondent, the sample proportion of domestic violence is substantially lower than the average from the rural sample of the ENDIREH data, in which 41% of rural Mexican women

¹²Program eligibility is based on poverty status. Therefore, I considered all poor households in both treatment and control villages. Only poor households in treatment villages are offered the treatment.

¹³The questions in the original language are: "Quien(es) es(son) la(s) persona(s) que mas bebe(n) de este hogar, aunque sea de vez en cuando?", and "Cuando bebe, esta persona tiene una actitud agresiva?"

 $^{^{14}\}mathrm{A}$ possible reason for the vague phrasing of the sentence may be its sensitive content.

 $^{^{15}}$ More precisely, 21% drink 1 to 3 times per month, 12.5% 1 to 4 times per weeks, and 2.5% daily or almost daily.

15 or older who live with a partner were victims of some form of violence in 2003, and 10% were victims of alcohol-induced violence. The under-reporting is consistent with evidence from different databases that victims of violence may not provide this type of information because they are afraid of their husband's reaction. Moreover, in some cases the husband may be at home while the wife is interviewed.¹⁶ Because of this under-reporting, the estimated effects of income on aggressive behavior may be a lower bound of the true effect.

Alcohol abuse and aggressive behavior increase with the household poverty level. In villages where the cash transfer to women is not offered, alcohol abuse and related aggressive behavior for the best off quartile are respectively 20% and 40% lower than for the remaining 75% of the sample. However, the evidence from the national surveys is more complex. For example, in the ENA2002 data alcohol abuses are more frequent in (wealthier) urban than (poorer) rural areas. Further, violence and female education follow an inverse u-shaped relationship: 38% of uneducated women sampled in ENDIREH are victims of violence; the proportion peaks at 52% for women with secondary education and declines beyond that point. Economically active women and women from rural areas are respectively 3 percentage point more likely and 5 percentage points less likely to be victims of violence than a randomly drawn woman. This evidence suggests that violence and alcohol abuse may be caused by multiple factors and confirms that the estimation of the causal effect of income may help unveil some determinants of these phenomena, as well as provide us with potential policy tools.

The other key variables in the analysis are wife and husband's income. The data on income are very detailed. For each household member aged 6 and above, I observe earnings and additional benefits from the main occupation, earnings from an additional job, net earnings from informal activities, income from pension, interests, rentals (land, machinery, and animals), and income from community-wide activities. I also know the value of state aid from various programs, including Oportunidades.¹⁷ After I aggregated these data for wives and husbands separately, I dropped the few missing observations and further trimmed the top percentile and the few negative values (fewer than 1%).¹⁸ Average wife and husband monthly incomes are 139 and 536 *pesos*, or roughly 14 and 54 dollars. While only 12% of husbands reported zero earnings, 48% of wives have no income. Oportunidades transfers are the major source of income for wives; their average income is respectively 16 and 217 *pesos* in control and treatment village.

¹⁶The document summarizing the evidence from ENDIREH 2003 provides interesting, though saddening anecdotal evidence of problems with disclosing sensitive information and husbands monitoring the wives' answers.

¹⁷These programs are Ninos de Solidaridad, INI (National Indigenous Institute), Empleo Temporal, PROBE-CAT, and PROCAMPO.

¹⁸Negative income levels are possible because the money invested for informal activities may exceed the revenues.

The income of both spouses is 35% higher after the wife receives the transfer.

Income is likely endogenous, as it may be correlated to unobservable determinants of alcohol abuse and violence, therefore I need instrumental variables to obtain consistent estimates of its effect. Since one such variables is the random assignment to Oportunidades, I need to briefly describe some relevant features of the program, the data, and the randomization. In November 1998, Oportunidades provided bimonthly grants to poor households in the form of income subsidies, as well as scholarships for children attending third to ninth grade, which corresponded approximately to one half to two thirds of the wage a child would earn by working full time (Schultz, 2004), and the average monthly transfer amounted to 204 *pesos*, about 20 dollars. In these data only a random group of 320 villages participates to the program, out of the total group of 506 localities. Importantly, the cash transfers are handed to women only; thus, the randomized treatment assignment to the program increases women's incomes without being correlated with any unobservable characteristic.

The instruments for household income are job loss due to health shocks occurring to household members besides the husband, natural disasters, and village male agricultural wages. 0.7% of households have at least one member (besides the head) who could not work because of temporary health problems. In the six months preceding the interview, 42% of households were hit by at least one of the following natural disasters: water shortage, flood, frost, fire, pest, earthquake, and hurricane.¹⁹ The share of households hit by at least one such natural disaster varies substantially both within and between villages, as shown in Figure 1. Agriculture is the main activity in 97% of villages; the daily wage averages 26.6 pesos and it varies substantially.²⁰

Lastly, I observe the village price for one liter of aguardiente, the local liquor, as reported by a village official. In the case of unavailable price information, I use average municipal or state price instead. I use these price variables as exclusion restrictions in a selection model that I will discuss in the next section. Table 1 provides the means and standard deviations of the key variables.

4 Identification and estimation

Consider the following structural model:

$$D_i = \alpha_0 + \alpha_1 Y_i^w + \alpha_2 Y_i^h + \alpha_3 X_i + u_i \tag{1}$$

 $^{^{19}\}mathrm{The}$ most frequent natural disaster was water shortage, which affected 37% of the sampled households.

 $^{^{20}}$ Its standard deviation is 9.2.

$$V_i = \beta_0 + \beta_1 Y_i^w + \beta_2 Y_i^h + \beta_3 X_i + \epsilon_i \quad \text{if} \quad D_i = 1$$

$$\tag{2}$$

$$V_i = 0 \quad \text{if} \quad D_i = 0 \tag{3}$$

$$Y_{i}^{j} = \delta_{0}^{j} + \delta_{1}^{j} Z_{i} + \delta_{2}^{j} X_{i} + v_{i}^{j} \quad \text{for} \quad j = \{w, h\}$$
(4)

where D is a dummy for alcohol abuse, V an indicator for alcohol-induced aggressive behavior, Y^w is the wife's income, and Y^h the husband's. Since both incomes are likely endogenous, I use the instrumental variables Z mentioned in the previous section: random assignment to Oportunidades, health-related employment shock of household members other than the husband, natural disasters, and male agricultural wages. The X variables are frequency of health checks, regional dummies, intensity of natural disasters at the locality level, and village poverty level. The variables u, ϵ , and v^j are the unobserved determinants of alcohol abuse, aggressive behavior, and incomes, respectively.

I am interested in the effect of husband and wife's incomes on husband's alcohol abuse and aggressive behavior. Their effect on alcohol abuse is captured by the parameters α_1 and α_2 . As regards the effect on aggressive behavior, I have two sets of parameters of interest: the effect of incomes on violence for both the individuals who drink and the whole sample. I will discuss identification and estimation of these parameters in turn.

4.1 The effect of income on alcohol abuse

The main identification issue in this case is the endogeneity of husband and wife's income. I assume I have valid instruments, that is i) Cov(Z, u) = 0, ii) the $\hat{\delta}_1^{\hat{w}}$ and $\hat{\delta}_1^{\hat{h}}$ are significant, and iii) the instruments are not weak. Since the zero covariance assumption is non-testable, I will first discuss whether the exclusion restrictions are plausible for each instrument.

The instrument for wife's income is program assignment. Since this assignment is random, the instrument is uncorrelated with pre-program unobservables by design. However, this variable may not be excludable, as it may affect the dependent variables besides its effect through income. There are four channels through which Oportunidades may affect alcohol abuse and aggressive behavior over and above its effect on female income: first, the program may influence alcoholism and violence also by causing an increase in school attendance. Women and children who were otherwise isolated may have contact with other people and report the abuses to the teachers; bruises and signs of violence may be noticed. To avoid this possibility, I dropped all families with at least one eligible child who was not enrolled in school in the academic year before the program started.²¹ Thus, I restrict the sample to households without eligible children

 $^{^{21}}$ I identify these children in the following way: first, they have to be up to 17 years old in September 1997.

or families whose eligible children were already attending school in the 1997-1998 academic year.

Second, program recipients are required to have quarterly health checks; the checks are respectively every three and two months for adults and children. Although infrequent, these checks may provide a disincentive for the husband to be physically abusive towards his wife and children. To control for this potential effect, I add dummies for frequency of health visits in the previous 6 months.²² It turns out that most households did have health checks irrespective of their eligibility to Oportunidades; 72% of households in control villages had health checks, against 86% in treatment villages.

Third, the program may affect prices. However, Angelucci and De Giorgi (2006) show that this is not the case. In any case, I compared alcohol prices of treatment and control villages, which showed that average aguardiente prices are 11.95 in control villages and 12.05 in treatment villages; their difference is not statistically significant. Since village-level data on alcohol price were available in only 118 of the total 506 sampled villages, I computed the average municipal price when the village price was missing.²³ I also repeated this exercise considering only villages for which price data were available. The average prices for this smaller sample are 10.56 and 9.71 in treatment and control villages, and again their difference is not significant. As a further check, I estimated equations (1) to (4) alternatively including and omitting alcohol price from the set of covariates X; the results do not change.

Fourth, the cash recipients have to attend nutrition and health classes. These classes may teach women that domestic violence is an unacceptable habit, provide them with an opportunity to voice their discontent, and give support for confronting their husbands. To test that this is not the case, I estimate the effect of Oportunidades on the households who did not qualify for the program (because they are not poor enough), but who live in treated villages. The women of those households were also strongly encouraged to attend the meetings; however, the program does not change their earnings.²⁴ I do not find any significant effect for this group,

This is because the program is only offered to individuals 18 and younger. Second, they must have completed grades 1 to 8 in the 1996-1997 academic year. Children with lower or higher education levels are not eligible for the program, which provides pupils attending grades 3 to 9 with scholarships in 1998 and 1999. I include in my sample all the families with no such children, as well as families with children in this category, as long as all of them were going to school in the academic year 1997-1998. For example, if one family has two children aged 9 and 10 with 3 years of education, but only one is enrolled in school in September 1997, I drop the entire family from my sample. This is because the eligibility to Oportunidades may induce the family to send the child to school, and this may affect the husband's drinking and abusive behavior in the ways described above. Note that families did not know about the program when they had to enroll their children to school in 1997.

²²The possible responses to whether the family had health checks in the previous 6 months are "yes"; "yes, at times"; "yes, rarely"; "no".

 $^{^{23}}$ In a few cases I used instead average state prices, because there were no alcohol data at the municipal level.

²⁴These households receive more transfers and loans from friends and relatives (Angelucci and De Giorgi, 2006). However, there is no difference in the amounts of loans and transfers to non-eligible wives in treatment

suggesting that this channel may not be important, or that it may operate only when there is also a positive income change.²⁵

Program assignment as an instrument identifies the expected change in alcohol abuse for each extra unit of income that wives eligible for Oportunidades have over wives who live in control villages, or:

$$\frac{E[D|T = 1, X, Y^h] - E[D|T = 0, X, Y^h]}{E[Y^w|T = 1, X, Y^h] - E[Y^w|T = 0, X, Y^h]}$$
(5)

where T is the random program assignment indicator (T=1 in Oportunidades villages, and 0 otherwise).²⁶ In the result section, I will report both the marginal effect and the effect of the average transfer size on the dependent variable, an estimate of the numerator of this ratio.

The three potential instruments for husband's income are the idiosyncratic component of natural disasters, village average daily agricultural wage, and employment shocks. The village aggregate component of natural disasters may affect drinking and abusive behaviors, for example by changing prices of labor, goods, and capital. Indeed, the average intensity of natural disasters at the village level is negatively and significantly correlated with both wages, by -28%, and alcohol price, by -22%. To control for this effect, I included the average incidence of natural disasters at the village level, a set of regional dummies, and the village poverty level to the set of conditioning variables, to capture both the long-term and the current riskiness of each locality.²⁷ The idea is that there are other factors besides this type of shock that influence prices, and I am assuming that these factors are conditionally independent of alcohol abuse and violence. Thus, my identification assumption is that in two villages from the same region and with the same current level of poverty and of natural disasters, the difference in wages and in whether the individual households suffered a shock will affect the dependent variables only through changes in income.

The case for the additional instrument, job loss of household members (other than the head) due to temporary incapacitation, is more straightforward: the identification assumption is that this type of health shock affects husband's alcohol abuse and aggressive behavior only by changing his income. Since only 0.7% of households have at least one such member, I can rule out price effects.

and control villages.

 $^{^{25}}$ I estimated a reduced-form model of alcohol abuse and alcohol-induced violent behavior on the treatment assignment dummy and the X covariates using data on the non-poor in treatment and control villages: the marginal effects (and standard errors) were -0.014 (0.018) and 0.003 (0.006), respectively.

 $^{^{26}}$ This is under the assumption that the other instruments do not have a significant effect on wife's income.

 $^{^{27}}$ The average incidence of natural disasters at the village level is the fraction of households who were hit by at least one natural disaster in the previous 6 months.

One may argue that more vulnerable households may be both more prone to health shocks or natural disasters and more likely to have alcoholic and abusive husbands than other households. This hypothesis is non-testable. However, if this were the case, one would expect past consumption and income to be lower for families currently hit by a shock than for families who are not. I regress September 1997 household monthly income and March 1998 monthly food expenditure per adult equivalent, which averaged respectively 758 and 603 *pesos* (at current prices), on dummies for whether each type of shock occurred between May and November 1998, conditioning on the X covariates. The estimated coefficients are neither statistically nor economically significant, as shown in Table 2. Lastly, these shocks possibly have psychologic effects that may induce alcoholism or violence. I assume that these effects are functions of income, e.g. that a head whose household suffered a shock without incurring any income loss will not change his drinking and aggressive behavior.

These instruments identify the effect of random income shocks due to such factors as a lower than average harvest (if wages depend on output), or a loss of manpower caused by a co-worker's illness (if relatives work together). Importantly, while the wife's income change is expected and somewhat long-lasting, the husband's income shock is unexpected, and probably short-termed, since future shocks do not predict past income and expenditure.²⁸

Although (1) is a linear probability model, I estimate it using both linear and non-linear estimators: 2SLS, LIML, and probit. When using linear probability models, I interpret the estimated coefficient as locally linear marginal effects at the mean of the explanatory variables. The same applies for the violence regressions.

4.2 The effect of income on alcohol-induced violence

I am interested in estimating two sets of parameters: the effect of income on the aggressive behavior of individuals who drink, and the overall effect of income on alcohol-related violence.

First, consider the effect of income on violence for drinkers. If income causes both alcohol abuse and aggressive behavior, an income increase will change both the set of individuals who drink and their likelihood of being violent. Therefore, the instrumental variables may no longer be independent of the unobservables in equation (2), since they select a different set of drinkers. In this case, I can identify the causal effect of income on violence under the following additional assumptions, besides the instrument validity assumptions i) to iii): iv) $u_i \sim N(0, 1)$; v) ϵ_i is a

²⁸The program characteristics are discussed in village assemblies; the transfers are initially guaranteed until the end on 1999.

linear function of u_i , $E(\epsilon_i | u_i) = \lambda u_i$. Consider the following selection equation:

$$D_{i} = 1\{\gamma_{0} + \gamma_{1}X_{i} + \gamma_{2}Z_{i} + u_{i} > 0\}$$

Under these assumptions, $E(\epsilon_i|X, Z, D = 1) = \lambda I$, where $I = \frac{\phi(\gamma_0 + \gamma_1 X_i + \gamma_2 Z_i)}{\Phi(\gamma_0 + \gamma_1 X_i + \gamma_2 Z_i)}$ is the inverse Mills ratio. The above assumptions enable me to re-write equation (2) as

$$V_{i} = \beta_{0} + \beta_{1} Y_{i}^{w} + \beta_{2} Y_{i}^{h} + \beta_{3} X_{i} + E(\epsilon_{i} | X, Z, D = 1) + e_{i}$$
(6)

where $e_i = \epsilon_i - E(\epsilon_i | X, Z, D = 1)$. I can obtain an estimate of the inverse Mills ratio using the parameters from a probit regression of D on X and Z. Since $E(e_i|X, Z, D = 1) = 0$ by construction, I can regress V on income, the X covariates and the estimated inverse Mills ratio by 2SLS, and compute the standard errors with the bootstrap. This is a selection model with endogenous explanatory variables, as described in Wooldridge (2002). Identification is achieved through both the non-linearity of the inverse Mills ratio and the use of variables that affect aggressive behavior only through drinking. The instruments Z include both the variables discussed above and village-level alcohol price data; these are the price of one liter of aguardiente, and a dummy for price availability at the village level, as reported by the interviewed official. The absence of information on alcohol price may either mean that this product is not sold locally (probably because there is no local store or market), or that the respondent is either a teetotaller or does not purchase aguardiente. The absence of a local liquor retail outlet likely increases the effective price of alcohol, as one has to add transportation costs. Further, a non-drinking elected official may represent local preferences against alcohol. Thus, my additional identification assumptions are that, conditional on observables, villages with higher alcohol prices or with missing alcohol price data are less likely to have alcohol-abusing husbands, but that a higher price and price availability have no direct effect on alcohol-induced violence.

The second parameters of interest are the effects of wife and husband's income on alcoholinduced aggressive behavior for the whole sample. The expected level of violence, irrespective of whether the husband drinks, is the following:

$$E[V|X, Y^{w}, Y^{h}] = E[V|X, Y^{w}, Y^{h}, D = 1] \times P(D = 1|X, Y^{w}, Y^{h})$$
(7)

Thus, the overall effect of income on aggressive behavior is the sum of two partial effects: the direct effect of income on aggressive behavior for drinkers, and its indirect effect through changes in alcohol abuse. This is a non-linear function of income, if both effects are statistically significantly.

5 Results

5.1 The effect of shocks and treatment assignment on wife and husband's income

Table 3 provides OLS estimates of the first-stage regression coefficients. Treatment assignment is the only significant instrument in the wife's income regression: wives in treatment villages have on average 195 more pesos than wives in control villages; husband's income does not differ by village type. Husband's income is a negative function of both natural disasters and job loss of household members; compared to the average income in a household without negative shocks, husband's income is 46 pesos lower in households that suffered from some natural disaster, and 279 pesos lower if any relative could not work because of temporary health issues (see column 3). This latter effect suggests that there may be complementarities in income production, such as those generated by relatives working on the same plot of land or having the same business. An alternative explanation for this lower income may instead be that health shocks are correlated, e.g. both the husband and another relative fall ill: however, out of the households with one such health shock, only in 15% of cases the husband is temporarily incapacitated too. Moreover, the coefficient is the same when I estimate this regression omitting households where the head also suffered a health shock.

Interestingly, the effect of temporary incapacitation of the wife on husband's income is significantly larger than the effect of other members' health shocks. In unreported regressions, I estimated the separate effect of wife and other household members' health shocks on husband's income, and found that the coefficients are respectively -376 and -184 pesos, both statistically different from zero and from each other at the 1% level. This is perhaps due to wives being more likely than other relatives to work in the same business as their husbands'. Alternatively, when the wife falls ill the husband may have to attend to some of her duties, having less time to do paid work. Lastly, there is a positive and significant effect of agricultural wages on husband's income; a one peso increase in daily wage is associated with a 4 peso higher husband income.

The F test of joint significance of the instruments and the partial R^2 are very high in the the wife's income first-stage regression, but less so for husband's income. Nevertheless, the value of the F test of joint significance of the instruments for husband's income is never less than 10, which is Staiger and Stock's (1997) rule of thumb to detect weak instruments.²⁹

5.2 The effect of wife and husband's income on alcohol abuse and aggressive behavior

Table 4 shows estimates of the effect of wife and husband's income on husband's alcohol abuse. Higher wife's income causes a significant significant decrease in the husband's alcohol abuse; a 200 peso increase in her monthly income (i.e. the approximate income increase caused by the program) reduces the likelihood of alcohol abuse by 4.6 percentage points, a 15% drop compared with control villages (columns 1 and 3). The 2SLS estimates are lower than the OLS one, which are 0.001 and not statistically different from zero (column 2). This suggests OLS is upward biased.³⁰ Changes in husband's income, on the other hand, have no significant effect on the likelihood of abusing alcohol consumption; the point estimates are positive, but very imprecisely estimated.

The estimated effects do not change when I instrument for one endogenous variable only (columns 4 and 5). When there are two potentially endogenous explanatory variables X_1 and X_2 , 2SLS provides consistent estimate of X_1 as long as the instrumental variable used is uncorrelated with X_2 (Ackerberg and Crawford, 2006). This is possible here because the instruments for husband's income - job loss due to temporary incapacitation, natural disasters, and agricultural wage - are not significantly related to wife's income, and vice versa. This is potentially useful especially to estimate the effect of husband's income, because I can exploit all the variation in wife's income to try and improve the precision of its estimate. However, the estimated effect is still insignificant. The instruments for both wife and husband's income are not weak, and the estimates' significance and magnitude are broadly unchanged when I use different estimators, as I discuss in the Appendix.

Table 5 provides estimates from selection models that use price of aguardiente and price availability as instruments for the likelihood of drinking.³¹ Neither income coefficient is significantly different from zero. Thus, income changes do not seem to affect alcohol-induced violence for individuals who drink. However, the partial effect of husband's income must be interpreted with caution, since the instruments for husband's income are weak in this smaller sample. For example, when I regress husband's income on agricultural wages, health-related job loss of other

 $^{^{29}\}mathrm{This}$ rule of thumb applies for single endogenous explanatory variables.

³⁰The OLS estimate of wife's income is also positive but not significantly different from zero when I estimate it for households in control villages only.

 $^{^{31}{\}rm The}$ partial effects of alcohol price and price availability are respectively -0.061 (standard error 0.017) and -0.003 (0.001).

members (excluding wife and husband), and the two alcohol price variables (as well as on the X variables), the F-test of joint significance of the instruments is only 5.55, as shown in column 5.³² In case of weak instruments, 2SLS is biased towards OLS. The evidence against a causal effect of wife's income is instead stronger.³³

As for alcohol abuse, a comparison of the OLS and the 2SLS estimates shows that failing to control for the endogeneity of income results in upward-biased estimates.

Some of the individuals in this sample are husbands who did not stop drinking despite their wife's higher income. If individuals whose drinking habits are the least sensitive to their wife's higher income are also the most violent, then the effect of wife's income on this group may be a lower bound of the effect in the whole sample, and higher wife's income may deter aggressive behavior in at least some individuals. I can indirectly test for this hypothesis comparing the frequency of aggressive behavior for drinkers in control and treatment villages. Treatment villages have proportionally fewer drinkers than control villages because Oportunidades increases wives' incomes, causing a drop in alcohol abuse, without changing husbands' incomes. If the individuals who drink despite their wives' higher income are the most violent, their incidence of violence should be higher than for drinkers in control villages. Instead, the difference in aggressive behavior is actually lower in control (14.3%) than in treatment villages (13.7%), and their difference is not statistically significant, either with and without conditioning on the set of X variables. This provides further evidence against a direct effect of wife's income on alcohol-induced violence.

These results suggest that the overall effect of wife's income on alcohol-induced violence in this sample is simply the partial effect of income on alcohol abuse, times the probability of alcohol abuse:

$$\begin{split} & \frac{E[V|X,Y^h,T=1]-E[V|X,Y^h,T=0]}{E[Y^w|X,Y^h,T=1]-E[Y^w|X,Y^h,T=0]} = \\ & \frac{E[D|X,Y^h,T=1]-E[D|X,Y^h,T=0]}{E[Y^w|X,Y^h,T=1]-E[Y^w|X,Y^h,T=0]} P(V=1|X,Y^h,Y^w,D=1) \end{split}$$

Under the restriction that the direct effect of income on violence is zero, I can obtain a reducedform estimate of the overall effect of wife's income on violence simply by regressing violence on income for the whole sample and accounting for the endogeneity of wife's income.

³²I dropped natural disasters, which now appear to be insignificant.

³³As in the previous case, I checked that these results are robust to using different estimators and instruments, as reported in the Appendix.

Table 6 shows that the estimates of wife's income are negative, significant, and robust to the choice of different estimators. The estimate of the marginal effect from column 4 is -0.047, significant at the 95% level, showing that a 200 peso increase in monthly income causes a reduction in alcohol-induced aggressive behavior of 1 percentage point. This a 21% drop, compared with the incidence of violence in control villages, which is 4.7%.

5.3 Interpretation: determinants of alcohol abuse and violence, wife utility, and policy implications

A 20 dollar increase in wife's income reduces alcohol abuse by 15% and this, in turns, causes a 21% drop in the related aggressive behavior. The estimated effect of income on violence is probably an underestimate of the true effect in the sample, as the actual level of violence is likely higher than reported. This is consistent with Bobonis *et al.* (2007), who, using different violence data, find that physical and sexual abuse against women is 30 percent lower in beneficiary than in control households. These effects are consistent with the intrinsic view that drinking and violence provide direct utility to the husband, while reducing the wife's wellbeing; therefore, the wife's higher income enables her to reduce the level of husband alcohol abuse and aggressive behavior. This is not surprising, considering that the wives in my sample have hardly any income, besides the program transfers, so the alcohol abuses and violence observed in areas where Oportunidades is not implemented are unlikely caused by a need to control the wives' monetary resources.

An alternative explanation is that the drop in alcohol abuse and violence are caused by the wife's more credible divorce threat, now that her income is higher, irrespective of whether the husband enjoys these goods *per se*, or uses them as tools to control the wife's income. The threat is probably not credible, though. Although the average Oportunidades transfer is more than a tenfold increase in wife's income, 200 *pesos* per month would not be sufficient to provide enough food for the wife and her children, if she were to leave her husband taking her children with her. The average monthly food consumption for adults and children are 128 and 94 *pesos* in the absence of the program (Angelucci and De Giorgi, 2006, and Di Maro, 2003). The wives in my sample have on average 1.14 children eligible for Progresa, and an additional 1.97 children aged 0 to 9, requiring in total 420 *pesos* per month only to provide food (including the food needed for the wife). Moreover, although guaranteed for two years, the transfers 1) may stop at the end of 1999, and 2) are linked to children's school attendance, and would eventually stop as the children complete the subsidized grades. Lastly, divorce is not as socially acceptable in Mexico as in other countries. For example, the divorce rate in rural Mexico is 0.10 per thousand

inhabitants, 5.7 times as small as the U.S. rate (United Nations Demographic Yearbooks 2002 and 2003).

How much money is the wife giving up to reduce the husband's alcohol abuse and drunken violence? While I have no quantitative data on intra-household transfers, I can look at how the wife's extra money is spent, and whether it crowds out transfers from the husband. For this purpose, I use follow-up interviews to the recipients of the Oportunidades grants, who explain how the program has affected them. These data are collected in May 1999 from the same sample of beneficiaries I observe in November 1998. Interestingly, only 5% of recipients receive fewer transfers from their husbands, and only 0.1% of recipients hand the money to their husbands. Instead, the wives manage the extra income, and transfer resources to the husbands by purchasing goods that increase the utility of all household members. The three most commonly purchased items are food (89%), shoes (38%), and clothes (33%), followed by school supplies, housewares, and medicines.³⁴ Thus, the cost of lower alcohol abuse and violence is only a fraction of the wife's extra income. For example, if the transfer is divided in equal shares between 5 household members, husbands on average receive resources worth 40 pesos per month.³⁵

The respondents have a positive perception of the program overall effects. When asked about how being an Oportunidades beneficiary affected their lives, the receipients' most common answers suggest increased freedom and security: 55% of women report feeling safer, 50% having money at their disposal, 20% being able to decide what to purchase, and 13% being free to spend on whatever they need. Only 0.1% report having problems with their spouses. This additional evidence suggests that husbands are not simply changing their type of violent behavior (e.g. from drunken to sober violence), but that the quantity of overall domestic violence must also be dropping.

These results have clear policy implications: providing poor women with opportunities to increase their income may reduce the husbands' alcohol dependence and drunken violence. One way to achieve this is by targeting women as direct recipients of cash transfers or of microfinance programs, both of which are becoming a common practice in developing countries. In order to assess the effectiveness of such policies, one would need some estimates of the cost of

³⁴The respondents could give multiple answers.

 $^{^{35}}$ These may actually be conservative estimates, because the household size of program beneficiaries is lightly bigger, and the largest increase in non-food expenditures if for child-specific goods: for instance, Attanasio and Lechene (2002) show that, as the share of household income brought by the wife increases, there is an increase in the consumption of food and child clothing, and a drop in tobacco and alcohol expenditures. Rubacalva *et al.* (2006) show that the income transferred to women through Oportunidades is spent on small livestock, on child goods, and to improve nutrition, and that overall it increases women's power in the household.

alcoholism and domestic violence, which, to my knowledge, are not available for Mexico.³⁶ At the very least, these findings lengthen the list of the known beneficial effects of cash transfers to poor women. At a broader level, they teach us that problems such as alcohol abuse and domestic violence, which seem to be endemic to particular cultures, are by no means ineradicable, and can be alleviated by an appropriate set of policies.

Another interesting result of this exercise is that alcohol-induced aggressive behavior declines only through a reduction in drinking; the higher wife's income does not reduce the aggressive behavior of drinkers. Therefore, when they are drunk husbands respond to incentives differently than when they are sober. This is not a new result: the medical and psychiatric literatures overwhelmingly point to this direction (e.g. Leonard, 2000, 2005). The importance of alcohol as a trigger for violence confirms that a further policy tool to reduce both alcohol abuse and violence may be alcohol price, consistent with the evidence provided by Grossman and Markovitz, and Mehlum *et al.* (2006).

Lastly, husband's income has no significant effect on alcohol abuse, while the instruments are not sufficiently strong to provide reliable evidence on its direct effect on domestic violence. Since the instruments are temporary income shocks with supposedly no long-term effects, these results suggest that alcohol abuse does not respond to transitory changes in husband's income. This behavior may instead be a function of permanent income as a proxy for the spouses' bargaining power, as shown by the fact that long-lasting changes in wife's income lower alcohol abuse.

6 Conclusions

Alcohol abuse and domestic violence are severe public health issues in Mexico; understanding their causes and how they respond to income changes is fundamental for their eradication. Using a sample of poor households from rural Mexico, I showed that long-lasting increases in wife's income reduce alcohol abuse and drunken aggressive behavior, unlike short-term fluctuations in husband's income, which have no significant effect.

I coupled the quantitative analysis with evidence that the higher wife's income is managed by the recipient, does not crowd out transfers from the husband, is spent on goods that increase

 $^{^{36}}$ The World Health Organization (WHO, 2004) reported an increase in mortality rates from ischaemic heart disease and cirrhosis of the liver, for which alcohol is one of the underlying risk factors: for example, liver cirrhosis is one of the top 10 causes of death among the Mexican population, and it is the most common cause of death among males between 35 and 54 years of age. The mortality rate due to alcohol has increased from 7.8/100 000 persons in 1970 to 12/100 000 persons in 1995 within the population 15 years of age and older (Medina-Mora *et al.*, 2000). Higher mortality is only a fraction of the overall costs of alcohol abuse and domestic violence.

the utility of all household members, and causes an increase in security and independence. These results have important policy implications, suggesting that female income or labor force participation may be used as policy tools to reduce alcoholism and domestic violence. They also confirm that the growing practice of targeting women as recipient of cash transfers or micro-credit has positive spillover effects in this dimension.

As usual, these results should be interpreted with caution: what applies for poor rural households may not work for wealthy urban ones. Further research is needed, for example, to assess the effect of income for wealthier households, for whom the instrumental value of alcohol abuse and domestic violence may be more relevant. A better understanding of these phenomena is essential for the design of appropriate policies: my paper is a step in that direction.

References

- [1] Ackerberg, D. and Crawford, G. (2006), "Estimating price elasticities in differentiated product demand models with endogenous characteristics", mimeo UCLA.
- [2] Angelucci, M. and De Giorgi, G. (2006), "Indirect effects of an aid program: How Do Liquidity Injections Affect Non-Eligibles Consumption?", mimeo, University of Arizona.
- [3] Attanasio, O. and Lechene, V. (2002), "Tests of Income Pooling in Household Decisions," *Review of Economic Dynamics*, Vol. 5(4), pp. 720-748.
- [4] Bloch, Francis and Vijayendra Rao (2002), "Terror as a Bargaining Instrument: A Case-Study of Dowry Violence in Rural India," American Economic Review, vol. 92(4), pages 1029-1043.
- [5] Blundell, R. and Smith, R. (1989), "Estimation in a Class of Simultaneous Equation Limited Dependent Variable Models", Review of Economic Studies, 56, 37-58.
- [6] Bobonis, G., Gonzalez-Brenes, M., and Castro, R. (2007), "Public transfers and spousal violence", mimeo, University of Toronto.
- [7] Bowlus, A and Seitz, S. (2006), "Domestic Violence, Employment and Divorce," International Economic Review, forthcoming.
- [8] Di Maro, V. (2004), "Evaluation of the Impact of Progress on Nutrition: Theory, Econometric Methods and an Approach to Deriving Individual Welfare Findings from Household Data", mimeo, University College London.

- [9] Duflo, E. and Udry, C. (2004), "Intrahousehold Resource Allocation in Cote d'Ivoire: Social Norms, Separate Accounts and Consumption Choices", NBER Working Paper no. 10498.
- [10] Encuesta Nacional de Adicciones (2002), Instituto Nacional de Estadstica, Geografa e Informica, Mexico.
- [11] Encuesta Nacional sobre la Dinmica de las Relaciones en los Hogares (2004), Instituto Nacional de Estadstica, Geografa e Informitca, Mexico.
- [12] Farmer, Amy and Jill Tiefenthaler (1996), "Domestic Violence: The Value of Services as Signals," American Economic Review, Vol. 86, No. 2, 274-279.
- [13] Farmer, Amy and Jill Tiefenthaler (1997), "An Economic Analysis of Domestic Violence," Review of Social Economy, Vol. 55, No. 3, 337-358.
- [14] Grossman, M. and Markowitz, S. (2000), "The Effects of Beer Taxes on Physical Child Abuse", Journal of Health Economics, 19(2), 271-282.
- [15] Grossman, M. and Markowitz, S. (2001), "Alcohol Regulation and Violence on College Campuses", in "The Economic Analysis of Substance Use and Abuse: The Experience of Developed Countries and Lessons for Developing Countries," edited by Michael Grossman and Chee-Ruey Hsieh, Edward Elgar Limited, United Kingdom.
- [16] Krug, E.G., Dahlberg, L.L., Mercy, J.A., Zwi, A.B., and Lozanoet R., eds. (2002), "World report on violence and health", World Health Organization, Geneva.
- [17] Leonard, K. E. (2000), "Domestic Violence and Alcohol: What is known and What do we need to know to Encourage Environmental Interventions," Paper Presented at Alcohol Policy 12, Washington, D.C., June 11-14, 2000.
- [18] Leonard, K. E. (2005), "Alcohol and intimate partner violence: when can we say that heavy drinking is a contributing cause of violence?", Addiction, 100(4), 422-25..
- [19] Ligon, Ethan, J.P. Thomas, and T. Worral (2002) "Informal insurance arrangements with limited commitment: Theory and evidence from village economies," The Review of Economic Studies, 69(1), 209-244.
- [20] Markowitz, S. (2000), "The Price of Alcohol, Wife Abuse and Husband Abuse", Southern Economic Journal, 67, No. 2, 279-303.

- [21] Markowitz, S. (2000), "The Role of Alcohol and Drug Consumption in Determining Physical Fights and Weapon Carrying by Teenagers," Eastern Economic Journal, 27, No.4 (Fall 2001), 409-432.
- [22] Markowitz, S. (2001), "An Economic Analysis of Alcohol, Drugs, and Violent Crime in the National Crime Victimization Survey", in *The Economic Analysis of Substance Use* and Abuse: The Experience of Developed Countries and Lessons for Developing Countries, edited by Michael Grossman and Chee-Ruey Hsieh, Edward Elgar Limited, United Kingdom.
- [23] Markowitz, S. and Grossman, M. (1998), "The Effect of Alcohol Regulation on Physical Child Abuse," NBER Working Paper, No. 6629.
- [24] Medina-Mora E., Borges G., Villatoro J. (2000), "The measurement of drinking patterns and consequences in Mexico", Journal of Substance Abuse, 2000, 12(12):183196.
- [25] Mehlum, H., Miguel, E., and Torvik, T. (2006), "Poverty and Crime in 19th Century Germany", forthcoming, Journal of Urban Economics.
- [26] Miguel, E. (2005), "Poverty and Witch Killing", Review of Economic Studies, 72(4), 1153-1172.
- [27] Morrison A.R., and Orlando M.B. (1999), "Social and economic costs of domestic violence: Chile and Nicaragua", in Morrison A.R., and Orlando B.E., eds. *Too close to home: domestic violence in the Americas* (51-80), New York, Inter- American Development Bank.
- [28] Pollak, Robert A. (2004), "An Intergenerational Model Of Domestic Violence," Journal of Population Economics, 2004, 17(2), 311-329.
- [29] Rubacalva, L., Teruel, G., and Thomas, D (2006), "Investments, time preferences and public transfers paid to women", mimeo UCLA.
- [30] Staiger, D. and Stock, J. H. (1997), 'Instrumental Variables Regression with Weak Instruments," Econometrica 65(3), 557-586.
- [31] Stevenson, B. and Wolfers, J. (2006), "Bargaining in the Shadow of the Law: Divorce Laws and Family Distress," Quarterly Journal of Economics, 121(1), 267-288.
- [32] Stock, J. H. and Yogo, M. (2005), "Testing for Weak Instruments in Linear IV Regression" (with James H. Stock), in D.W.K. Andrews and J.H. Stock, eds., *Identification and*

Inference for Econometric Models: Essays in Honor of Thomas Rothenberg. Cambridge: Cambridge University Press, pp. 80108.

- [33] Tauchen H. V., Ann Dryden Witte, and Sharon K. Long (1991), "Domestic Violence: A Nonrandom Affair", International Economic Review, Vol. 32, No. 2, pp. 491-511.
- [34] Tauchen, Helen V. and Ann Dryden Witte (1995), "The Dynamics of Domestic Violence," American Economic Review, Vol. 85, No. 2, 414-418.
- [35] United Nations (2005), "2002 Demographic Yearbook", Department of economic and social affairs, New York.
- [36] United Nations (2006), "2003 Demographic Yearbook", Department of economic and social affairs, New York.
- [37] Wooldridge, J. (2002) "Econometric analysis of cross section and panel data", the MIT Press: Cambridge, MA.
- [38] World Health Organization (2004), "Global Status Report on Alcohol 2004", Department of Mental Health and Substance Abuse, Geneva.
- [39] World Health Organization (2005), "WHO multi-country study on womens health and domestic violence against women: summary report of initial results on prevalence, health outcomes and womens responses," Geneva.

7 Appendix: robustness checks

Table 7 estimates the effects of income on alcohol abuse and violence using the same data described in the text, but trying different estimators and changing the set of instrumental variables. The magnitude and significance of the estimated effects do not vary. I report LIML estimates, which are robust in case of weak instruments, probit estimates, and 2SLS estimates accounting for the endogeneity of husband's income only. When I estimated equations (1) and (4) jointly, the likelihood function did not converge easily. Thus, I am reporting the results from a two-step procedure using a control function approach (e.g. as in Blundell and Smith 1989), estimating the standard errors with the block-bootstrap (where the block is the village). Columns 3 and 6 show that the instruments for husband's income are not weak; the Cragg-Donald F-test for weak instruments is 24.03. These values reject the null hypothesis of weak

instruments at the 5% level for a 10% desired maximal size of the Wald test.³⁷ Also, note that the overidentification tests can never reject the null that the instruments are uncorrelated with the error term, and that they are correctly excluded from the estimated equation.³⁸

My second set of robustness checks regards the effect of income on the unconditional likelihood of violence. First, I estimate the overall effect of income on alcohol-induced violence in two steps, multiplying the estimate of the partial effect of income on drinking by the average probability of violence in my sample, assuming that the partial effects of wife and husband's income on violence are both zero, and computing the standard errors by the bootstrap. In that case I obtained a point estimate of -0.033 (standard error 0.006) for the effect of wife's income, and of 0.008 (0.020) for the effect of husband's income.³⁹

Second, I measure the indirect effect of wife's income on violence by estimating the effect of drinking on the related violence, π_1 in the following regression,

$$V_i = \pi_0 + \pi_1 D_i + \pi_2 X_i + \nu_i$$

I then multiply this number by the effect of income on drinking from Table 4. Since drinking and violence may be both driven by unobserved factors, I use price of aguardiente, and the availability of this price, as instruments for drinking in the violence equation. The instruments are strongly significant in the first stage and pass both the weak instruments tests in Stock and Yogo (2005) and the overidentification test with flying colors.⁴⁰ The estimated partial effect of alcohol abuse on aggressive behavior is 0.320 (with a standard error of 0.089). Using the partial effect of wife's income on alcohol abuse from column 2 in Table 4, and estimating the standard errors with the bootstrap (selecting villages as the sampling block and having 500 repetitions), I estimate an indirect effect of wife's income on aggressive behavior of -0.074 (standard error 0.037).

My final set of checks controls that the results are robust to using a different subset of households, a different definition of income, and different instruments. First, I additionally excluded families whose children may finish primary school and have to choose whether to enroll in secondary school while the program is being offered, i.e. children who completed either fifth or sixth grade by the end of the academic year 1996-1997. These are the families whose

³⁷The critical values are 16.38 and 19.93 for a test based on the desired maximal size of a 5% Wald test of the significance of the 2SLS estimate for one endogenous variable and using respectively one and two instruments (Stock and Yogo, 2005).

 $^{^{38}\}mathrm{For}$ 2SLS, this test is Hansen's J statistic, for LIML it is the Anderson-Rubin statistic.

 $^{^{39}\}mathrm{I}$ used 500 replication and randomly selected blocks of villages.

⁴⁰The Cragg-Donald F-statistic is 21.41, the p-value of the overidentification test is 0.444.

children's schooling decisions are most likely affected by Oportunidades, since pre-program enrollment rates drop dramatically from primary to secondary school (i.e. from grade 6 to 7), and Oportunidades has been shown to be especially successful in furthering school attendance beyond sixth grade (columns 1-3 of Table 8).⁴¹ Second, I experimented with a different definition of income that includes the value of loans and transfers received by the wife and husband (columns 4-6 of Table 8). Third, I slightly changed the instrument set. For example, I replaced the health-induced job loss dummy with the number of household members who were ill in the previous month. The results are robust to these changes.

A further concern is that I did not detect any significant effect of husband's income on alcohol abuse and aggressive behavior because the sample size is too small. For this reason, I estimated the effect of husband's income on these two outcomes on the entire sample, including households non eligible for the program who live in control and treatment villages. In this way the overall sample sizes are 17312 and 4910. When I use the whole sample, the instruments are robust and the main results are unchanged; temporary shocks to husband's income do not affect alcohol abuse and violent behavior. However, the instruments are weak for the sample of drinkers (columns 7-9 of Table 8).

 $^{^{41}}$ However, note that the enrollment to the 1997-1998 academic year is most likely exogenous, because in September 1997 the households did not know about the program, which started in 1998.



Figure 1: Village share of households hit by at least one natural disaster between May and November 1998.

<u>Table 1: M</u>	<u>feans and sta</u>	<u>ndard dev</u>	<u>viations of t</u>	<u>he main variables</u>
Alcohol	Aggressive	Wife	Husband	Oportunidades
abuse	behavior	income	income	eligibility
0.29	0.04	0.14	0.54	0.61
(0.45)	(0.20)	(0.21)	(0.74)	(0.49)
Job loss	Natural	Daily	Price of	Price
(health)	disaster	wage	alcohol	unavailability
0.007	0.42	27.31	12.02	0.77
(0.08)	(0.49)	(10.00)	(4.80)	(0.42)

Wife and husband monthly incomes are in thousand *pesos*. Daily wages (in *pesos*) are average wages for male agricultural workers. No agricultural wage is reported in 19 villages. In that case, average municipal or state price are used instead. Wage and price data averages are based on 506 observations, the number of villages in the sample.

	Sept. 1997	March 1998
	income	food expenditure
Natural disaster	-7.72	-1.297
	[16.092]	[1.943]
Job loss (health)	28.661	-10.025
	[99.696]	[7.299]

Table 2: Difference in past income and consumption by current shock type

Standard errors clustered at the village level. The additional covariates are regional dummies, average natural disaster at the village level, village marginalization index, and frequency of health checks.

		ine bilocits (m whe and	nusbanu s i	
	(1)	(2)	(3)	(4)	(5)
	Wife	Wife	Husband	Husband	Husband
	income	income	income	income	income
Oportunidades	0.195	0.195	-0.025	-0.024	
	$[0.006]^{***}$	$[0.006]^{***}$	[0.024]	[0.024]	
Job loss (health)	0.01		-0.277		
(no husband)	[0.025]		$[0.042]^{***}$		
Job loss (health)		0.032		-0.181	
(no husband or wife)		[0.037]		$[0.061]^{***}$	
Natural disaster	0.001	0.001	-0.052	-0.053	
	[0.004]	[0.004]	$[0.016]^{***}$	$[0.016]^{***}$	
Agricultural wage ^{a}	0.005	0.006	0.432	0.430	0.444
	[0.047]	[0.046]	$[0.128]^{***}$	$[0.128]^{***}$	$[0.124]^{***}$
R^2	0.23	0.23	0.03	0.03	0.03
	Instrument	al variables'	partial R^2 :		
	0.20	0.20	0.004	0.003	0.002
	F test of jo	int significan	ice of the inst	trumental va	riables:
All	248.25^{***}	248.26^{***}	20.43^{***}	12.02^{***}	12.71^{***}
Excluding Oportunidades	0.10	0.35	26.47^{***}	14.18^{***}	12.71^{***}

Table 3: Effect of different income shocks on wife and husband's income

Standard errors clustered at the village level. ***, **, *: significant at the 1%, 5%, and 10% level. The additional regressors are frequency of visits to health clinics; village poverty level, regional dummies, and village intensity of natural disasters in the previous 6 months. ^a: the wage is in hundreds of pesos.

	(1)	(2)	(3)	(4)	(5)
	OLS	OLS	2SLS	2SLS	2SLS
	reduced form			wife only	husband only
Oportunidades	-0.046				
	$[0.016]^{***}$				
Job loss (health)	-0.033				
(no husband)	[0.051]				
Natural disaster	0.015				
	[0.081]				
Agricultural wage	0.006				
	[0.011]				
Wife income		0.001	-0.232	-0.234	0.002
		[0.028]	$[0.083]^{***}$	$[0.082]^{***}$	[0.028]
Husband income		0.001	0.018	0.001	0.029
		[0.006]	[0.124]	[0.006]	[0.119]
Instruments:					
Treatment assignm	ent		Yes	No	No
Agricultural wage			Yes	Yes	Yes
Natural disaster			Yes	No	Yes
Job loss (health) (r	no husband)		Yes	No	Yes
Job loss (health) (r	no husband/wife)	No	No	No
Overidentification t	test (p-value):		0.699		0.709
Cragg-Donald test	- /		10.74	2576.50	14.49

Table 4: Effect of wife and husband's income on the probability of alcohol abuse

Monthly income in thousand *pesos*. Standard errors clustered at the village level. ***, **, *: significant at the 1%, 5%, and 10% level. The sample size is 10257. The estimated effects of husband's income in column 4 and wife's income in column 5 are biased and inconsistent.

	(1)	(2)	(3)	(4)
	OLS	OLS	2SLS	2SLS
	Reduced form			wife only
Oportunidades	-0.004			
	[0.015]			
Agricultural wage	0.047			
	[0.081]			
Job loss (health)	0.274			
(no husband/wife)	$[0.136]^{**}$			
Wife income		0.093	0.105	0.085
		$[0.035]^{***}$	[0.127]	[0.095]
Husband income		-0.009	-0.035	-0.011
		[0.08]	[0.185]	[0.008]
Inverse Mills ratio	-0.277	-0.294	-0.300	-0.276
	[0.117]**	$[0.099]^{***}$	$[0.167]^*$	$[0.125]^{**}$
Instruments:				
Treatment assignment	No	No	Yes	Yes
Agricultural wage	No	No	Yes	No
Job loss (health) (no husband/wife)	No	No	Yes	No
Alcohol price	Yes	Yes	Yes	Yes
Price availability	Yes	Yes	Yes	Yes
F stats of instruments in 1st stage:				
Wife income			100.43^{***}	165.91^{***}
Husband income (all IVs)			5.55^{***}	
Husband income (no price			11.15***	
or treatment assignment)				

Table 5: Effect of wife and husband's income on the probability of aggressive behavior conditional on drinking

Monthly income in thousand *pesos*. Standard errors clustered at the village level. ***, **, *: significant at the 1%, 5%, and 10% level. In case of block-bootstrap estimates of the standard errors, the block is the village. The sample size is 2936. The estimated effect of husband's income in column 4 is biased and inconsistent.

	(1)	(2)	(3)	(4)	(5)
	OLS	OLS	2SLS	2SLS	2SLS
	reduced form			wife only	husband only
Oportunidades	-0.009				
	[0.005]*				
Natural disaster	0.015				
	[0.027]				
Agricultural wage	0				
	[0.004]				
Wife income		0.017	-0.043	-0.047	
		[0.011]	$[0.026]^*$	$[0.024]^{**}$	
Husband income		-0.003	0.023	-0.003	0.027
		[0.002]	[0.050]	[0.002]	[0.050]
Instruments:					
Treatment assignm	ent		Yes	No	No
Agricultural wage			Yes	Yes	Yes
Natural disaster			Yes	No	Yes
Job loss (health) (r	o husband)		No	No	No
Job loss (health) (r	no husband/wife)	No	No	No
Overidentification (test (p-value):		0.676		0.640
Cragg-Donald test	. ,		11.01	2576.50	16.79

Table 6: Effect of wife and husband's income on the unconditional probability of aggressive behavior

Monthly income in thousand *pesos*. Standard errors clustered at the village level. ***, **, *: significant at the 1%, 5%, and 10% level. These regressions do not use the health-induced job loss variables as instruments because I rejected the exogeneity test for these variables (I performed a C statistic, i.e. the difference of the Hansen-Sargan statistic of the equation with the smaller set of instruments - valid under both the null and alternative hypotheses - and the equation that includes the job loss instrument). The sample size is 10257. The estimated effects of husband's income in column 4 and wife's income in column 5 are biased and inconsistent.

	(1)	(2)	(3)	(4)	(5)	(6)
		Alcohol			Violenc	e
	LIML	IV-Probit	2SLS	LIML	IV-Probit	2SLS
			Husband only			Husband only
Wife income	-0.232	-0.232	0.001	-0.043	-0.044	0.018
	$[0.083]^{***}$	$[0.070]^{***}$	[0.028]	$[0.026]^*$	[0.028]	[0.011]
Husband income	0.019	0.018	0.006	0.022	-0.009	0.041
	[0.126]	[0.120]	[0.139]	[0.050]	[0.023]	[0.062]
Instruments:						
Treatment assignment	Yes	Yes	No	Yes	Yes	No
Agricultural wage	Yes	Yes	Yes	No	Yes	Yes
Natural disaster	Yes	Yes	Yes	No	Yes	No
Job shock (health)	Yes	Yes	No	No	No	No
(no husband)						
Overidentification test:	0.692			0.677		
(p-value)						
Cragg-Donald test	10.74		24.03	11.01		24.03

Table 7: Effect of wife and husband's income on alcohol abuse and (unconditional) aggressive behavior: using different estimators and instruments

Monthly income in thousand *pesos*. Standard errors clustered at the village level. ***, **, *: significant at the 1%, 5%, and 10% level. In case of block-bootstrap estimates of the standard errors, the block is the village. The estimated effects of wife's income are biased and inconsistent in columns 3 and 6.

																	rap estimates	
	umple	ome only)	(6)	Violence	(unconditional)	2SLS			-0.004	[0.079]			17312			9.43	In case of block-bootst	
income	Larger se	usband inc	(8)	Violence	(if D=1)	2SLS			-0.137	[0.293]	-0.130	[0.213]	4910			2.07	1 10% level.	
finition of		(h	(2)	Alcohol		2SLS			0.066	[0.206]			17312			9.43	1%, 5%, and	
nple and the de	transfers		(9)	Violence	(unconditional)	2SLS	-0.051	$[0.028]^{*}$	-0.002	[0.052]			10257		242.53	12.09	significant at the	
ing the san	ne including		(5)	Violence	(if $D=1$)	2SLS	0.107	[0.117]	-0.025	[0.143]	-0.301	$[0.141]^{**}$	2936		100.32	5.31	ol. ***, ** *. . , *	
hecks: vary	Incor		(4)	Alcohol		2SLS	-0.236	$[0.085]^{***}$	0.018	[0.122]			10257	essions:	242.97	20.34	he village leve	
3: Robustness c	1/6th graders		(3)	Violence	unconditional	2SLS	-0.08	$[0.039]^{**}$	0.02	[0.064]			8095	ie first-stage regr	285.09	11.13	rors clustered at t	
Table 8	with no 5th		(2)	Violence	(if D=1)	2SLS	-0.047	[0.194]	-0.156	[0.209]	-0.228	[0.184]	2255	ficance in th	83.75	4.84	Standard er	the village.
	Families		(1)	Alcohol		2SLS	-0.252	$[0.117]^{**}$	0.027	[0.136]			8095	joint signi	213.76	21.88	usand <i>pesos</i> .	the block is
							Wife income		Husband income		Inverse Mills ratio		Observations	F-test of instrument	Wife income	Husband income	Monthly income in tho	of the standard errors,